Author’s response to reviews

Title: GABA-B Limbic Encephalitis and Asystolic Cardiac Arrest: A case report

Authors:
Chris Ovens (ovens.chris@gmail.com)
Angelo Jayamanne (angelojayamanne@gmail.com)
Andrew Duggins (andrew.duggins@sydney.edu.au)

Version: 2 Date: 18 Nov 2017

Author's response to reviews:

Thankyou for your review and comments. In response to your issues raised:
1) Was the patient physical injured?
- He suffered minor soft tissue injuries but no significant trauma.

2) Present detailed physical and neurological examination after his accident as well as his first admission. What was the temperature, pulse, blood pressure and temperature, on admission? Give all results of laboratory findings (i.e. CBC, liver and renal functions), etc)
- Unfortunately we have had significant difficulty in obtaining further documentation from the hospital which the patient initially presented to. All examination and investigations that we have has been included. We are happy to include further results as they are made available.

3) In the Discussion – Start with summarizing the case and describe what is unique in this case compared to what is available in the literature.
- This has been included.